Author’s response to reviews

Title: A comprehensive evaluation of the role of genetic variation in follicular lymphoma survival

Authors:

Fredrik Baecklund (fredrik.baecklund@ki.se)
Jia-Nee Foo (foojn@gis.a-star.edu.sg)
Paige Bracci (paige.bracci@ucsf.edu)
Hatef Darabi (hatef.darabi@ki.se)
Robert Karlsson (robert.karlsson@ki.se)
Henrik Hjalgrim (hhj@ssi.dk)
Richard Rosenquist (richard.rosenquist@igp.uu.se)
Hans-Olov Adami (hadami@hsph.harvard.edu)
Bengt Glimelius (bengt.glimelius@onkologi.uu.se)
Mads Melbye (mme@ssi.dk)
Lucia Conde (lconde@uab.edu)
Jianjun Liu (liuj3@gis.a-star.edu.sg)
Keith Humphreys (keith.humphreys@ki.se)
Christine F Skibola (cskibola@uab.edu)
Karin E Smedby (karin.ekstrom.smedby@ki.se)

Version: 4
Date: 24 September 2014

Author’s response to reviews: see over
Dear Editor,

We were very pleased to receive your message about the acceptance of our manuscript “A comprehensive evaluation of the role of genetic variation in follicular lymphoma survival”. We have now made the changes requested by the Editors, for details, please see below.

Sincerely,

Fredrik Baecklund, on the behalf of all authors
Stockholm, September 24th 2014
EDITORIAL REQUIREMENTS:

Please revise your ethics statement to include the names of the ethics committees that approved your study.

The names of all ethics committees have now been added in the ethics statement (page 7, rows 20-12)

Please provide email address details for all authors on the title page of the manuscript.

The email addresses of all authors are now provided on the title page of the manuscript.